Author's response to reviews

Title: Carbohydrate mouth rinse: does it improve endurance exercise performance?

Authors:

Vitor de Salles Painelli (vitor.painelli@usp.br)
Humberto Nicastro (nicastroh@yahoo.com.br)
Antonio H Lancha Jr (lanchajr@usp.br)

Version: 4 Date: 24 August 2010

Author's response to reviews:

The changes to the format of the manuscript were done. A corrected version is being submitted.